communications medicine

ARTICLE

https://doi.org/10.1038/s43856-023-00335-4

Check for updates

Association of birth and childhood weight with risk of chronic diseases and multimorbidity in adulthood

OPEN

Yue Zhang[®] ^{1,2,7}, Yaguan Zhou^{1,2,7}, Yangyang Cheng^{1,2}, Rodrigo M. Carrillo-Larco³, Muhammad Fawad[®] ^{1,2}, Shu Chen^{4,5} & Xiaolin Xu[®] ^{1,2,6⊠}

Abstract

Background Little is known about the relationship between early life body size and occurrence of life-course multiple chronic diseases (multimorbidity). We aim to evaluate associations of birth weight, childhood body size, and their changes with the risks of chronic diseases and multimorbidity.

Methods This prospective cohort study included 246,495 UK Biobank participants (aged 40-69 years) who reported birth weight and childhood body size at 10 years old. Birth weight was categorized into low, normal, and high; childhood body size was reported as being thinner, average, or plumper. Multimorbidity was defined as having two or more of 38 chronic conditions retrieved from inpatient hospital data until 31 December, 2020. The Cox regression and quasi-Poisson mixed effects models were used to estimate the associations.

Results We show that 57,071 (23.2%) participants develop multimorbidity. Low birth weight (hazard ratio [HR] 1.29, 95% confidence interval [CI] 1.26-1.33), high birth weight (HR 1.02, 95% CI > 1.00-1.05), thinner (HR 1.21, 95% CI 1.18-1.23) and plumper body size (HR 1.06, 95% CI 1.04-1.09) are associated with higher risks of multimorbidity. A U-shaped relation-ship between birth weight and multimorbidity is observed. Changing to be thinner or plumper is associated with multimorbidity and many conditions, compared to changing to be average. **Conclusions** Low birth weight, being thinner and changing to have a thinner body size in childhood are associated with higher risks of developing multimorbidity and many chronic conditions in adulthood. Early monitoring and maintaining a normal body size in childhood could have life-course benefits for preventing multimorbidity above and beyond individual conditions.

Plain language summary

Little is known about the relationship between childhood body size and the risk of developing more than one chronic disease later in life. Using data from the UK, we found that low birth weight, high birth weight, and being thinner or plumper than average during childhood were all associated with higher risks of developing more than one chronic disease in adulthood. In addition, changing body shape during childhood to be either thinner or plumper, was associated with being more likely to develop more than one chronic disease later in life. Our results highlight the importance of early monitoring and maintenance of average body size in childhood, as this might prevent the occurrence of chronic diseases later in life.

¹School of Public Health and The Second Affiliated Hospital, Zhejiang University School of Medicine, Hangzhou, Zhejiang, China. ² Key Laboratory of Intelligent Preventive Medicine of Zhejiang Province, Hangzhou, Zhejiang, China. ³ Hubert Department of Global Health, Rollins School of Public Health, Emory University, Atlanta, GA, USA. ⁴ Australian Research Council Centre of Excellence in Population Ageing Research (CEPAR), University of New South Wales, Sydney, Australia. ⁵ School of Risk & Actuarial Studies, University of New South Wales, Sydney, Australia. ⁶ School of Public Health, Faculty of Medicine, The University of Queensland, Brisbane, Australia. ⁷These authors contributed equally: Yue Zhang, Yaguan Zhou. ¹²

he World Health Organization has endorsed a global target of a 30% reduction in low birth weight and no increase in the proportion of overweight children by 2025¹. Birth weight is an indicator of infant intrauterine development and health², and current studies have also associated it with incidences of chronic diseases in adulthood or later life. A cohort study using data from 256,787 participants of UK Biobank observed that participants with LBW had higher risks of coronary heart disease, stroke, and heart failure in their later life³. Another study of 15,792 individuals aged 45-64 years in the Atherosclerosis Risk in Communities Study found that high birth weight (HBW) was related to an increased risk of heart failure⁴. Childhood body size was also reported to associate with a series of chronic conditions, including obesity, diabetes, metabolic syndrome, depression, and cancer⁵⁻¹⁰. However, evidence on whether and to what extent abnormal birth weight and childhood body size are associated with life-course risks of multiple chronic conditions, which is defined as multimorbidity¹¹, remains scanty.

Most previous studies mainly focused on a single measurement of birth weight or body size at a specific childhood phase, failing to assess the dynamic weight changes over time. An observational study combined five birth cohorts measured the weight gain in three age periods (0-2 years, 2 years to mid-childhood, and midchildhood to adulthood) and found faster weight gain was associated with an increased risk of overweight and high blood pressure in adulthood¹². Another cohort study reported an association between weight gain from birth to age 2 years and lung function in late life¹³. These findings endorsed the notion that childhood developmental differences are evident in the first few years after birth, and worsen during early childhood, resulting in continuous influence on health throughout life¹⁴. However, most current studies commonly measured weight gain or weight growth, ignoring the influence of shifting from a normal or plumper body size to a thinner body size. As an indicator of undernutrition or developmental delays, children with thinner body sizes were more likely to develop chronic conditions related to nutrition in later life¹⁵. Understanding the long-term consequences of different changing patterns of body size in early childhood is important for life-course monitoring and intervention of body weight and weight-related chronic diseases.

As birth weight and childhood body size are manageable^{16–18}, elucidating their life-course association with multimorbidity and chronic conditions would have implications for preventing health-related outcomes around the whole life-course. Leveraging data from 246,495 individuals from UK Biobank, our study aims to investigate the association of birth weight, childhood body size at 10 years, and their longitudinal changes with the late-life risk of 38 types of chronic conditions and multimorbidity. In our study, birth weight and childhood body size were associated with multimorbidity and many chronic conditions in later life. Individuals who changed body shape during childhood to be either thinner or plumper, no matter from which birth weight group, were more likely to develop more than one chronic disease later in life.

Methods

Study design and participants. This cohort study used UK Biobank data and followed the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) reporting guideline. The UK Biobank is a large population-based prospective study, recruiting more than half a million UK individuals aged 40–69 years between 2006 and 2010. At baseline, sociodemographic, lifestyles, and health-related information were collected via touchscreen questionnaires and physical measurements. Participants were followed for health-related outcomes through linkage to routinely available national datasets, including

the Hospital Episode Statistics for England, the Patient Episode Database for Wales, and the Scottish Morbidity Record for Scotland. Appropriate informed consent was obtained from participants, and ethical approval was covered by the UK Biobank. More details of the UK Biobank study are available elsewhere¹⁹.

This study included individuals who attended the baseline assessment of UK Biobank. The exclusion criteria were: (1) individuals with no information on birth weight, childhood body size, or adulthood body mass index (BMI); (2) individuals who were part of a multiple birth; and (3) individuals who were lost or died during follow-up (Supplementary Fig. S1).

Assessment of birth weight, childhood body size, and weight change. The main exposures were birth weight and childhood body size at 10 years of each eligible participants, which were collected through self-reported questionnaires at baseline. Participants were asked to recall their birth weights and were categorized into three groups: LBW (≤ 2.5 kg), normal birth weight (NBW, 2.5–4.0 kg), and HBW (≥ 4.0 kg)^{20,21}. Childhood body size was self-reported according to the question: "When you were 10 years old, compared to average would you describe yourself as...?", with the answer of being thinner, average, or plumper. Furthermore, we assessed the weight change from birth to childhood and classified weight change into 9 groups.

Assessment of chronic conditions and multimorbidity. Based on previous literature, including the Quality and Outcomes Framework (QoF, with the common conditions in the UK)²², a large UK-based study²³, a systematic review on multimorbidity indices²⁴, and other evidence from UK Biobank on multimorbidity²³, a total of 38 physical and mental chronic conditions were included in our study to define multimorbidity (Supplementary Table S1). Information of aforementioned conditions was extracted through the linkage to inpatient hospital data, according to the codes of International Classification of Diseases 10th Revision (ICD-10). The onset time of each condition was recorded as the date of inpatient diagnosis. Multimorbidity was defined as the coexistence of two or more of these 38 chronic conditions.

Covariates. Covariates including sociodemographic characteristics (age, sex, ethnicity, Townsend deprivation index, and education levels), anthropometric information (BMI at baseline), lifestyle factors (current smoking and drinking status, physical activity, and intake of fruits and vegetables), and early life factors (maternal smoking status around birth and breastfeed status as a baby) were collected at baseline. Participants with missing data for covariates were assigned to a separate group of 'unknown'. Townsend deprivation index was divided into tertiles (low, moderate, and high). Adulthood BMI was categorized into 4 groups of underweight (<20.0 kg/m²), normal weight $(20.0-24.9 \text{ kg/m}^2)$, overweight $(25.0-29.9 \text{ kg/m}^2)$, and obesity $(\geq 30.0 \text{ kg/m}^2)^{25}$. Physical activity was evaluated via Metabolic Equivalent Task (MET) minutes per week for moderate and vigorous activity and was further categorized into tertiles (low, moderate, and high). For the intake of fruits and vegetables, participants were categorized into four groups respectively, according to previous studies²⁶⁻²⁸: '<2.0 servings/day', '2.0-2.9 servings/day', '3.0-3.9 servings/day', and '≥4.0 servings/ day'.

Identification of multimorbidity patterns. Exploratory factor analysis was performed with the principal factor method to identify multimorbidity patterns. The Kaiser-Meyer-Olkin statistic, estimated to analyze the adequacy of the sample, was 0.74. The breakpoint of the screen plot, the eigenvalue (>1), and interpretability were used to determine multimorbidity patterns. We performed a varimax rotation of factor-loading matrices, with the resulting factor loadings representing the strength of the association between each chronic condition and the latent pattern. Chronic conditions with a factor loading > 0.35 referred to the characteristics of the multimorbidity pattern. Factor scores of all participants were calculated to indicate adherence to each multimorbidity pattern, as higher scores represented higher adherence. They were then divided into the percentage of 0–85%, 85–90%, 90–95%, and 95–100% to indicate adherence to each multimorbidity pattern for further analysis.

Statistics and reproducibility. Basic characteristics were described as mean (standard deviation [SD]) for continuous variables and as number (percentage) for categorical variables. The differences of variables across groups by birth weight and childhood body size were compared using one-way analysis of variance (ANOVA) and χ^2 test.

In the primary analyses, Cox regression models were utilized to estimate the associations of birth weight and childhood body size with each chronic condition and multimorbidity. Years from birth to the onset of multimorbidity or each chronic condition, or 31 December 2020, whichever came first, were considered the time scale. The proportional hazard assumption was checked and verified using Schoenfeld's residual methods, where no violations were found. Four models were conducted to sequentially adjust covariates, including sociodemographic characteristics and anthropometric information, lifestyle factors, and early life factors. The restricted cubic spline (RCS) model was utilized to display the dose-response association of birth weight with multimorbidity and each chronic condition. Also, the cumulative incidence function (CIF) curves were used to compare the cumulative incidence of adulthood multimorbidity across different groups of birth weight and childhood body size. Moreover, associations of weight change with the incidence of each chronic condition and multimorbidity were assessed in each birth weight group using Cox proportional hazard models, with changes to average body size as reference groups. Besides, quasi-Poisson mixed effects models were used to evaluate the associations of birth weight, childhood body size, and weight change with the number of chronic conditions during follow-up.

A series of exploratory analyses, sensitivity analyses and subgroup analyses were conducted to test the robustness of our results. First, multinomial logistic regression models were used to explore the association of birth weight and childhood body size with different multimorbidity patterns. Second, we recategorized birth weight into four groups: ≤2.5 kg, 2.5-4.0 kg, 4.0-4.5 kg, and >4.5 kg^{29,30}, to explore the association of highest birth weight (>4.5 kg) with the risk of multimorbidity. Third, we repeated the primary analyses by: (1) excluding participants with missing data of covariates; (2) imputing missing data of covariates using multiple imputation for five times; and (3) including family history (mother/father/sibling) of several major diseases in the fully adjusted model: heart disease, stroke, cancer, chronic bronchitis/emphysema, high blood pressure, diabetes, Alzheimer's disease/dementia, Parkinson's disease, and severe depression (Supplementary Table S2). Fourth, subgroup analyses were performed stratified by age at baseline, sex, Townsend deprivation index, education level, and family history of major diseases. Potential effect modifications were evaluated by adding an interaction term of each variable and birth weight or childhood body size in the model.

Statistical analyses were performed using SAS (version 9.4, SAS Institute Inc., NC, USA) and R (version 4.1.2, R Foundation for Statistical Computing). Hazard ratios (HRs), odds ratios (ORs), relative risks (RRs) and 95% confidence interval (95% CI) were reported in this study. Significance tests were evaluated at the 0.05 level using two-sided tests.

Ethical consideration. The UK Biobank has ethical approval from the NHS National Research Ethics Service (16/NW/0274). This is a secondary analysis of UK Biobank which is exempted from Institutional Review Board.

Reporting summary. Further information on research design is available in the Nature Portfolio Reporting Summary linked to this article.

Results

Basic characteristics of participants. Among the 502,490 participants attending baseline assessment of UK Biobank, a total of 246,495 participants remained in the current study according to the inclusion and exclusion criteria. The basic characteristics of excluded and included participants were summarized in Supplementary Table S3. Among the 246,495 participants (Supplementary Data 1), the mean baseline age was 55.2 (SD = 8.1) years, and 150,486 (61.1%) participants were female. The proportion of participants who reported their birth weight was 77.2% (NBW), 8.8% (LBW), and 14.0% (HBW), respectively. A total of 51.0% of participants reported an average childhood body size at 10 years, whereas 32.3% reported being slimmer and 16.6% reported being plumper. The characteristics of participants according to birth weight and body size at 10 years were also shown in Supplementary Data 1.

A total of 57,071 (23.2%) participants developed multimorbidity. The characteristics of participants by multimorbidity status were presented in Supplementary Data 2. Participants with multimorbidity were more likely to have abnormal birth weight (LBW or HBW) and childhood body size (thinner or plumper), higher BMI at baseline, and be physically inactive than those free of multimorbidity.

Birth weight, childhood body size, and late-life multimorbidity and individual chronic conditions. Participants with LBW/ HBW and thinner/plumper childhood body size had a higher cumulative incidence of multimorbidity (Fig. 1). LBW (adjusted HR = 1.29, 95% CI = 1.26–1.33), HBW (adjusted HR = 1.02, 95% CI = > 1.00–1.05), thinner body size (adjusted HR = 1.21, 95% CI = 1.18–1.23), and plumper body size (adjusted HR = 1.21, 95% CI = 1.04–1.09) were all associated with a higher risk of developing multimorbidity (Table 1). The RCS model showed a U-shaped relationship between birth weight and multimorbidity (*P* for nonlinear <0.001, Fig. 2), with a lower risk at a birth weight of 3.33–4.09 kg. As shown in Table 2, low birth weight, high birth weight, thinner body size, and plumper body size were associated with increased number of chronic diseases.

Associations of birth weight and childhood body size with latelife individual chronic conditions were shown in Fig. 3. LBW and thinner body size was associated with higher risk of many chronic conditions. However, the results for HBW and plumper body size diverged. The U-shape relationship was also observed for birth weight with most of the 38 conditions (Supplementary Fig. S2–4).

Weight change from birth to childhood and late life multimorbidity and individual chronic conditions. Among participants with LBW, 43.2% kept being thinner at 10 years, and 15.4% became plumper. More than half of participants (51.6%) with



Fig. 1 Cumulative incidence of multimorbidity according to birth weight and childhood body size. For the curve of birth weight, the red, blue and green lines represent low birth weight, high birth weight, and normal birth weight, respectively. For the curve of childhood body size, the red, blue, and green lines represent thinner, plumper, and average body size, respectively.

NBW changed to be thinner at their 10 years. For participants with HBW, 26.1% of them were of average body size at 10 years, and 52.7% of them turned out to be thinner (Table 1). Changing to be thinner or plumper, no matter from which birth weight group, was observed to associate with higher risk of multimorbidity and increased number of chronic conditions, compared to changing to be average. The higher HRs occurred in changing to thinner body size (HR = 1.21, 95% CI = 1.15-1.28 for LBW to thinner body size; HR = 1.20, 95% CI = 1.18-1.23 for NBW to thinner body size; HR = 1.21, 95% CI = 1.15-1.27 for HBW to thinner body size). Also, changing to thinner or plumper body sizes were all associated with increased number of chronic conditions (Table 2).

The associations of weight change from birth to childhood with the risks of each chronic condition were showed in Fig. 4. For those with NBW, changing to be thinner at 10 years increased the risks of 28 conditions, whereas changing to be plumper was associated with higher risks of 12 conditions and lower risks of two conditions, respectively.

Additional analyses. According to factor analysis, we identified six patterns of multimorbidity (Supplementary Table S4). The associations of LBW and thinner body size with multimorbidity patterns were consistently observed, while plumper body size was associated with a lower risk of Cardiovascular Pattern (OR = 0.94, 95% CI = 0.89-0.99) (Supplementary Table S5). After recategorization of high birth weight, participants with the highest birth weight (>4.5 kg) had a higher risk of multimorbidity (adjusted HR = 1.10, 95% CI = 1.06-1.14), while that for participants with birth weight of 4.0-4.5 kg was not significant (Supplementary Table S6). Similar results were revealed after excluding or imputing the missing data of covariates (Supplementary Table S7-8). When additionally adjusting family history of several major disease in the model, similar results were found (Supplementary Table S9). The subgroup analyses showed similar results to our main findings (Supplementary Fig. S5). Moreover, the associations of birth weight and childhood body size with multimorbidity were consistent among participants with or without family history of most major diseases (Supplementary Fig. S6).

Discussion

In this large prospective cohort study of 246,495 participants, we found that birth weight and childhood body size were significantly associated with multimorbidity and many chronic conditions in late life. A U-shape relationship was observed between birth weight and multimorbidity and most chronic conditions. Individuals who changed to be thinner or plumper in childhood, no matter from which birth weight group, had increased risks of developing multimorbidity and many chronic conditions than those who changed to the average body size.

Previous cohort studies have explored the association between early life body size and several individual chronic conditions, mainly focusing on cardiometabolic diseases and cancers. Most of these studies have shown significant associations of birth weight and childhood body size with chronic conditions. For example, existing evidence has presented positive associations of LBW with higher risks of diabetes, metabolic syndrome, and cardiovascular disease (CVD)^{3,31-33}, and HBW with increased risks of obesity and schizophrenia^{34,35}. Ahlgren et al. also found a positive linear relationship between birth weight and many types of cancers³⁶. For body size in childhood, a higher BMI was observed to increase the risk of diabetes^{37,38}. A systematic review and metaanalysis of 37 longitudinal studies also found that higher childhood BMI could increase the risk of diabetes, coronary heart disease and cancers, but not stroke or breast cancer⁷. Although the prevalence of multimorbidity has been increasing rapidly in recent decades, particularly among the aging population, whether it could be predicted by early life factors has not been well documented. Our study extends previous findings to a broader spectrum of physical and mental conditions to measure multimorbidity in a large study sample with long-term monitoring of health-related outcomes, showing associations of LBW, HBW, thinner and plumper body size with higher risks of 27, 11, 27, and 13 of the 38 chronic conditions respectively. Birth weight was an indicator of fetal health^{2,39}, and our results confirmed its

Table 1 Associations of birth weight and childhood body size

with multimorbidity.							
Categories of birth weight, childhood body size, and weight change							
Birth weight							
	Low birth weight	Normal birth weight	High birth weight				
n (%)	21,626 (8.8)	190,359 (77.2)	34,510 (14.0)				
Model 1	1.32 (1.28-1.35)	Reference	1.05 (1.02-1.07)				
Model 2	1.29 (1.26-1.33)	Reference	1.02 (>1.00-1.05)				
Model 3	1.29 (1.26-1.33)	Reference	1.02 (>1.00-1.04)				
Model 4	1.29 (1.26-1.33)	Reference	1.02 (>1.00-1.05)				
Childhood	d body size						
	Thinner body size	Average body size	Plumper body size				
n (%)	79,700 (32.3)	125,803 (51.0)	40,992 (16.6)				
Model 1	1.22 (1.20-1.24)	Reference	1.05 (1.02-1.07)				
Model 2	1.21 (1.18-1.23)	Reference	1.06 (1.04-1.09)				
Model 3	1.21 (1.18-1.23)	Reference	1.06 (1.04-1.09)				
Model 4	1.21 (1.18-1.23)	Reference	1.06 (1.04-1.09)				
Groups o	f weight change						
	Low birth	Low birth	Low birth				
	weight \rightarrow Thinner	weight→Average	weight→Plumper				
n (%)	9,340 (43.2)	8,950 (41.4)	3,336 (15.4)				
Model 1	1.24 (1.17-1.31)	Reference	1.13 (1.06-1.22)				
Model 2	1.21 (1.15-1.28)	Reference	1.13 (1.05-1.21)				
Model 3	1.21 (1.15-1.28)	Reference	1.13 (1.05-1.21)				
Model 4	1.21 (1.15-1.28)	Reference	1.13 (1.05-1.22)				
	Normal birth	Normal birth	Normal birth				
	weight \rightarrow Thinner	weight→Average	weight→Plumper				
n (%)	98,293 (51.6)	61,740 (32.4)	30,326 (15.9)				
Model 1	1.22 (1.19-1.24)	Reference	1.03 (1.00-1.06)				
Model 2	1.20 (1.18-1.23)	Reference	1.05 (1.02-1.08)				
Model 3	1.20 (1.18-1.23)	Reference	1.05 (1.02-1.07)				
Model 4	1.20 (1.18-1.23)	Reference	1.05 (1.02-1.07)				
	High birth	High birth	High birth				
	weight→Thinner	weight→Average	weight→Plumper				
n (%)	18,170 (52.7)	9,010 (26.1)	7,330 (21.2)				
Model 1	1.24 (1.18-1.30)	Reference	1.08 (1.02-1.14)				
Model 2	1.21 (1.15-1.27)	Reference	1.08 (1.03-1.14)				
Model 3	1.21 (1.15-1.27)	Reference	1.08 (1.03-1.14)				
Model 4	1.21 (1.15-1.27)	Reference	1.08 (1.03-1.14)				
Model 1: unadjusted.							

Model 2: adjusted for age at baseline, sex, ethnicity, Townsend deprivation index, education levels, and BMI at baseline.

Model 3: model 2 also adjusted for current smoking and drinking status, physical activity, intake of fruits and vegetables.

Model 4: model 3 also adjusted for maternal smoking around birth and breastfed as a baby. We adjusted for the other variable in each of the birth weight and childhood obesity status model.

Hazard ratios and 95% confidence intervals are shown in the table.

association with multimorbidity and many chronic conditions in late life, especially LBW. In addition, the consideration of childhood body size further expanded the exploration of early life risk factors to a 10-year range. Our findings can support the "fetal origins" and DOHaD hypotheses that the disproportionate growth in early life, in utero in particular, would determine the development of many chronic diseases in late life⁴⁰, with substantial implications for individuals, families, societies, and the global healthcare systems.

It is worth noting that individuals with HBW were found to experience lower risks of asthma, diabetes, hypertension, and endometriosis, according to our results. The RCS models found the U-shape dose-response relationships between birth weight and three of these conditions (asthma, diabetes, and hypertension), and the lower HRs occurred in higher birth weight (>4.0 kg) than those for other conditions, which might explain the inverse association of HBW with these three conditions. We further conducted a sensitivity analysis by recategorizing HBW



Fig. 2 Restricted cubic spline models for the relationship of birth weight with and multimorbidity. Adjusted for age at baseline, sex, ethnicity, Townsend deprivation index, education level, smoking status, drinking status, physical activity, intake of fruits and vegetables, body mass index categories at baseline, maternal smoking around birth and breastfed as a baby. The solid line represents hazard ratios, and the shading represents 95% confidence interval.

Table 2 Association of birth weight, childhood body size, and weight change with the number of chronic conditions.						
Categories of birth weight, childhood body size, and weight change	n (%)	RR (95% CI)				
Birth weight						
Low birth weight	21,626 (8.8)	1.27 (1.25-1.29)				
Normal birth weight	190,359 (77.2)	Reference				
High birth weight	34,510 (14.0)	1.05 (1.03-1.06)				
Childhood body size						
Thinner body size	79,700 (32.3)	1.18 (1.16-1.19)				
Average body size	125,803 (51.0)	Reference				
Plumper body size	40,992 (16.6)	1.02 (1.01-1.03)				
Groups of weight change						
Low birth weight→Thinner	9340 (43.2)	1.18 (1.13-1.22)				
Low birth weight→Average	8950 (41.4)	Reference				
Low birth weight→Plumper	3336 (15.4)	1.09 (1.03-1.14)				
Normal birth weight→Thinner	98,293 (51.6)	1.17 (1.15-1.19)				
Normal birth weight→Average	61,740 (32.4)	Reference				
Normal birth weight→Plumper	30,326 (15.9)	1.01 (0.99-1.02)				
High birth weight→Thinner	18,170 (52.7)	1.20 (1.16-1.24)				
High birth weight→Average	9010 (26.1)	Reference				
High birth weight→Plumper	7330 (21.2)	1.04 (1.01-1.08)				
Adjusted for age at baseline, sex, ethnicity, To	ownsend deprivation inc	lex, education level,				

Adjusted for age at baseline, sex, ethnicity, Townsend deprivation index, education level, smoking status, drinking status, physical activity, intake of fruits and vegetables, BMI categories at baseline, maternal smoking around birth and breastfed as a baby. Relative risks and 95% confidence intervals are shown in the table.

into 4.0–4.5 kg and >4.5 kg groups and found the significant association with multimorbidity disappeared in the 4.0–4.5 kg group, while that for the highest birth weight (>4.5 kg) group remained. These results suggested that highest birth weight may be a better indicator of multimorbidity and several diseases, which might also partially explain the inconsistency with previous evidence (e.g., a previous study suggested the higher risk of

	Birth	Birth weight		Childhood body size		
	Low birth weight	High birth weight	Thinner	Plumper		
Atrial fibrillation	0.95 (0.77-1.16)	1.01 (0.87-1.17)	1.10 (0.97-1.25)	1.20 (1.02-1.40)		
Angina –	1.38 (1.29-1.47)	1.19 (1.12-1.26)	1.27 (1.21-1.33)	0.95 (0.90-1.01)		
Anxiety –	1.24 (1.14-1.36)	1.00 (0.93-1.08)	1.20 (1.13-1.27)	1.07 (0.99-1.15)		
Asthma-	1.31 (1.25-1.37)	0.93 (0.89-0.97)	1.33 (1.28-1.37)	1.14 (1.10-1.19)		
Bronchiectasis –	1.23 (1.02-1.50)	1.15 (0.97-1.35)	1.23 (1.08-1.40)	0.93 (0.78-1.11)		
Cancer-	1.05 (1.00-1.09)	1.11 (1.07-1.15)	1.01 (0.98-1.03)	0.91 (0.88-0.94)		
Cirrhosis –	1.75 (1.30-2.35)	1.04 (0.77-1.40)	1.00 (0.79-1.27)	0.93 (0.71-1.23)		
Chronic kidney disease –	1.42 (1.27-1.58)	0.97 (0.88-1.08)	1.16 (1.07-1.26)	0.97 (0.88-1.08)		
Chronic obstructive pulmonary disease	1.68 (1.53-1.84)	1.18 (1.08-1.29)	1.26 (1.18-1.35)	1.12 (1.03-1.23)		
Dementia –	1.28 (0.90-1.81)	1.02 (0.73-1.42)	1.17 (0.91-1.51)	1.12 (0.81-1.54)		
Depression –	1.24 (1.15-1.33)	0.95 (0.89-1.01)	1.25 (1.18-1.31)	1.25 (1.17-1.32)		
Diabetes –	1.49 (1.41-1.58)	0.94 (0.89-0.99)	1.38 (1.33-1.44)	1.12 (1.06-1.17)		
Eczema or dermatitis –	1.04 (0.90-1.20)	1.08 (0.96-1.21)	1.20 (1.09-1.31)	1.01 (0.90-1.14)		
Epilepsy-	1.34 (1.16-1.54)	1.11 (0.98-1.26)	1.09 (0.99-1.21)	0.96 (0.85-1.09)		
Glaucoma –	1.30 (1.16-1.45)	1.13 (1.02-1.24)	1.03 (0.95-1.11)	0.99 (0.90-1.10)		
Hepatitis –	2.16 (1.11-4.18)	1.76 (0.98-3.17)	0.96 (0.54-1.70)	1.59 (0.89-2.84)		
Heart failure –	1.30 (1.14-1.47)	1.21 (1.09-1.35)	1.21 (1.10-1.32)	1.04 (0.93-1.16)		
Hypertension –	1.32 (1.28-1.36)	0.96 (0.93-0.98)	1.19 (1.17-1.22)	1.04 (1.01-1.07)		
Irritable bowel syndrome	1.35 (1.23-1.48)	0.94 (0.86-1.04)	1.28 (1.20-1.37)	1.11 (1.02-1.21)		
Inflammatory bowel disease –	1.06 (0.93-1.20)	1.11 (1.00-1.23)	1.15 (1.06-1.25)	0.96 (0.87-1.08)		
Myocardial infarction –	1.21 (1.09-1.34)	1.14 (1.04-1.24)	1.17 (1.09-1.25)	0.89 (0.81-0.97)		
Migraine –	1.07 (0.93-1.23)	0.96 (0.85-1.08)	1.23 (1.12-1.35)	1.08 (0.96-1.21)		
Multiple sclerosis –	1.25 (1.00-1.57)	0.99 (0.81-1.21)	0.87 (0.74-1.02)	1.34 (1.12-1.60)		
Osteoporosis -	1.47 (1.31-1.64)	1.09 (0.97-1.22)	1.05 (0.97-1.15)	1.09 (0.97-1.22)		
Parkinson's disease –	1.14 (0.84-1.54)	1.29 (1.02-1.63)	1.11 (0.91-1.35)	0.91 (0.70-1.18)		
Prostate problem-	0.91 (0.83-1.01)	1.45 (1.36-1.55)	1.16 (1.09-1.23)	0.82 (0.76-0.89)		
Peripheral vascular disease –	1.49 (1.30-1.71)	1.05 (0.92-1.20)	1.03 (0.92-1.14)	1.32 (1.17-1.49)		
Rheumatoid arthritis –	1.11 (0.95-1.30)	0.96 (0.83-1.10)	1.20 (1.08-1.34)	1.06 (0.93-1.22)		
Schizophrenia –	1.25 (0.85-1.83)	1.06 (0.75-1.49)	1.73 (1.32-2.26)	1.28 (0.91-1.79)		
Chronic sinusitis -	1.17 (1.00-1.38)	1.06 (0.92-1.22)	1.21 (1.08-1.35)	1.08 (0.94-1.24)		
Stroke -	1.30 (1.13-1.50)	1.05 (0.92-1.19)	1.23 (1.11-1.35)	0.99 (0.88-1.13)		
Thyroid problem -	1.37 (1.30-1.45)	0.95 (0.90-1.00)	1.10 (1.06-1.15)	1.10 (1.04-1.15)		
Dyspepsia –	1.30 (1.21-1.39)	1.01 (0.95-1.08)	1.12 (1.06-1.18)	1.05 (0.99-1.12)		
Constipation-	1.28 (1.19-1.39)	1.07 (0.99-1.14)	1.17 (1.11-1.24)	1.11 (1.04-1.19)		
Hearing loss –	1.38 (1.22-1.56)	1.26 (1.13-1.40)	1.15 (1.05-1.25)	0.95 (0.85-1.07)		
Diverticular disease of intestine –	1.12 (1.06-1.17)	1.04 (0.99-1.08)	1.19 (1.15-1.23)	1.00 (0.96-1.04)		
Endometriosis –	1.19 (1.06-1.34)	0.72 (0.64-0.81)	1.19 (1.10-1.28)	1.29 (1.17-1.42)		
Meniere's disease –	1.25 (0.91-1.72)	1.37 (1.06-1.77)	1.43 (1.15-1.78)	1.17 (0.89-1.53)		

Fig. 3 Estimated hazard ratios (95% confidence intervals) for the association of birth weight and childhood body size with incident individual

conditions. Reference groups: normal birth weight and average body size in childhood. Adjusted for age at baseline, sex, ethnicity, Townsend deprivation index, education level, smoking status, drinking status, physical activity, intake of fruits and vegetables, body mass index categories at baseline, maternal smoking around birth and breastfed as a baby. Diseases with hazard ratios>1 and P < 0.05 were marked in red, those with hazard ratios<1 and P < 0.05 were marked in blue.

diabetes in those with HBW⁴¹). Besides, the negative linear relationship between birth weight and endometriosis has not been reported previously, calling for further clarification of such an association. In addition, the inverse association of plumper body size at 10 years with cancer, myocardial infarction, and prostate problems were observed in our study. Also, the inverse association of plumper body size with myocardial infarction might contribute to the inverse association between plumper body size and Cardiovascular Pattern. Consistent with our findings, the association of plumper body size in childhood could diverge with different subtype cancers, according to a previous study⁴². These discrepancies could partly be explained by the "growth acceleration" hypothesis that the catch-up growth for individuals with LBW or thinner childhood body size would result in relatively higher postnatal or pubertal growth and consequently lead to a higher risk of several conditions than those with HBW or plumper body size⁴³. However, due to the lack of continuous data on childhood body size, we could not find the cut-off of childhood BMI on multimorbidity and chronic conditions. Further large population-based studies are warranted to validate the role of HBW and plumper body size in childhood on the observed associations.

A previous study of UK Biobank observed a significant interaction between birth weight and adult BMI³, indicating that the longitudinal weight change might relate to long-term health status. Therefore, we further investigated the life-course association of weight change from birth to childhood with multimorbidity and chronic conditions. We found the association of changing to be thinner or plumper with multimorbidity was robust, regardless of individuals' birth weight. In line with our findings, a systematic review reported that individuals changing from normal weight to high weight in adulthood would experience higher CVD risk factors and outcomes⁴⁴, supporting that unfavorable weight change may adversely impact subsequent health-related outcomes. However, no previous study has well explored the effect of weight change from birth to childhood. As a proxy indicator of early life growth, more attention should be paid to exploring the association of weight change from birth to

	Low birth weight		Normal birth weight		High birth weight		
		↓					
	Thinner	Plumper	Thinner	Plumper	Thinner	Plumper	
Atrial fibrillation	1.07 (0.58-1.99)	1.53 (0.71-3.30)	1.14 (0.98-1.33)	1.26 (1.05-1.52)	0.99 (0.70-1.38)	1.06 (0.72-1.57)	
Angina	1.30 (1.14-1.48)	0.99 (0.83-1.18)	1.28 (1.21-1.35)	0.94 (0.88-1.01)	1.17 (1.04-1.32)	0.96 (0.84-1.09)	
Anxiety –	1.24 (1.04-1.48)	1.07 (0.84-1.36)	1.21 (1.13-1.30)	1.04 (0.95-1.14)	1.07 (0.90-1.27)	1.17 (0.98-1.40)	
Asthma –	1.29 (1.17-1.42)	1.22 (1.07-1.38)	1.31 (1.27-1.37)	1.12 (1.06-1.17)	1.43 (1.30-1.58)	1.22 (1.10-1.35)	
Bronchiectasis –	1.07 (0.72-1.60)	1.01 (0.58-1.75)	1.22 (1.05-1.42)	0.91 (0.73-1.12)	1.47 (1.04-2.07)	0.96 (0.62-1.50)	
Cancer-	1.08 (0.99-1.18)	0.95 (0.84-1.08)	0.99 (0.96-1.03)	0.90 (0.86-0.94)	1.02 (0.94-1.10)	0.93 (0.86-1.01)	
Cirrhosis -	0.58 (0.31-1.06)	0.41 (0.17-0.99)	1.11 (0.84-1.47)	1.07 (0.77-1.48)	1.09 (0.57-2.09)	1.00 (0.50-1.99)	
Chronic kidney disease –	1.20 (0.96-1.51)	1.23 (0.94-1.63)	1.12 (1.02-1.23)	0.92 (0.81-1.04)	1.44 (1.14-1.81)	1.01 (0.79-1.30)	
Chronic obstructive pulmonary disease	1.20 (1.00-1.43)	1.07 (0.84-1.37)	1.27 (1.16-1.38)	1.13 (1.02-1.25)	1.32 (1.10-1.60)	1.12 (0.91-1.37)	
Dementia –	1.94 (0.96-3.89)	0.45 (0.12-1.65)	0.93 (0.69-1.26)	1.17 (0.81-1.69)	1.81 (0.84-3.88)	1.62 (0.68-3.82)	
Depression -	1.17 (1.01-1.36)	1.47 (1.23-1.76)	1.26 (1.19-1.33)	1.23 (1.15-1.31)	1.25 (1.09-1.44)	1.20 (1.03-1.40)	
Diabetes -	1.33 (1.19-1.50)	1.23 (1.07-1.41)	1.39 (1.32-1.46)	1.08 (1.02-1.15)	1.41 (1.24-1.59)	1.19 (1.06-1.35)	
Eczema or dermatitis -	1.09 (0.80-1.47)	1.17 (0.79-1.73)	1.23 (1.10-1.36)	1.00 (0.87-1.15)	1.11 (0.86-1.43)	0.99 (0.75-1.31)	
Epilepsy-	1.14 (0.85-1.53)	1.36 (0.95-1.96)	1.12 (1.00-1.26)	0.96 (0.83-1.12)	0.95 (0.73-1.25)	0.77 (0.56-1.04)	
Glaucoma –	0.87 (0.69-1.10)	1.06 (0.78-1.44)	1.03 (0.94-1.13)	1.02 (0.90-1.15)	1.21 (0.99-1.48)	0.85 (0.67-1.09)	
Hepatitis-	0.45 (0.09-2.34)	1.75 (0.45-6.75)	0.69 (0.32-1.49)	1.72 (0.86-3.47)	3.14 (1.02-9.68)	0.95 (0.18-5.02)	
Heart failure –	0.99 (0.75-1.31)	1.25 (0.90-1.72)	1.28 (1.15-1.43)	1.01 (0.88-1.16)	0.99 (0.78-1.26)	1.00 (0.78-1.27)	
Hypertension -	1.16 (1.09-1.23)	1.09 (1.01-1.18)	1.19 (1.16-1.22)	1.02 (0.99-1.05)	1.24 (1.17-1.31)	1.09 (1.03-1.16)	
Irritable bowel syndrome-	1.31 (1.08-1.59)	1.18 (0.90-1.53)	1.30 (1.20-1.40)	1.12 (1.01-1.24)	1.18 (0.97-1.44)	1.03 (0.83-1.29)	
Inflammatory bowel disease -	1.07 (0.82-1.39)	1.07 (0.74-1.53)	1.18 (1.08-1.30)	0.98 (0.86-1.12)	1.05 (0.84-1.31)	0.84 (0.65-1.09)	
Myocardial infarction-	1.14 (0.93-1.40)	0.87 (0.64-1.17)	1.18 (1.08-1.27)	0.90 (0.81-1.00)	1.13 (0.94-1.35)	0.84 (0.69-1.04)	
Migraine -	1.38 (1.04-1.83)	0.89 (0.59-1.36)	1.22 (1.10-1.35)	1.15 (1.01-1.31)	1.18 (0.91-1.52)	0.85 (0.63-1.15)	
Multiple sclerosis-	1.25 (0.78-2.02)	1.57 (0.86-2.86)	0.83 (0.69-1.00)	1.23 (1.00-1.53)	0.78 (0.47-1.28)	1.78 (1.16-2.74)	
Osteoporosis -	1.13 (0.89-1.42)	1.23 (0.89-1.69)	1.05 (0.95-1.16)	1.01 (0.88-1.16)	1.01 (0.79-1.30)	1.32 (1.02-1.71)	
Parkinson's disease -	1.22 (0.64-2.34)	1.02 (0.42-2.50)	1.07 (0.85-1.34)	0.84 (0.61-1.15)	1.17 (0.71-1.94)	1.03 (0.59-1.80)	
Prostate problem-	1.08 (0.89-1.32)	0.63 (0.46-0.87)	1.16 (1.09-1.24)	0.82 (0.75-0.90)	1.17 (1.03-1.34)	0.89 (0.76-1.04)	
Peripheral vascular disease –	0.95 (0.71-1.26)	1.06 (0.73-1.53)	1.01 (0.89-1.14)	1.34 (1.17-1.55)	1.18 (0.88-1.59)	1.41 (1.04-1.89)	
Rheumatoid arthritis –	1.24 (0.89-1.72)	0.91 (0.56-1.45)	1.19 (1.06-1.35)	1.02 (0.87-1.20)	1.17 (0.86-1.61)	1.34 (0.98-1.85)	
Schizophrenia -	1.43 (0.64-3.21)	1.11 (0.41-3.02)	1.74 (1.27-2.37)	1.37 (0.92-2.03)	1.95 (0.96-3.95)	1.07 (0.45-2.54)	
Chronic sinusitis-	1.04 (0.74-1.45)	1.16 (0.74-1.81)	1.27 (1.12-1.44)	1.05 (0.89-1.24)	1.02 (0.75-1.40)	1.12 (0.81-1.54)	
Stroke-	1.29 (0.96-1.72)	1.23 (0.84-1.81)	1.25 (1.11-1.40)	0.94 (0.81-1.09)	1.03 (0.78-1.37)	1.05 (0.78-1.40)	
Thyroid problem-	1.26 (1.12-1.42)	1.25 (1.08-1.45)	1.07 (1.02-1.12)	1.07 (1.01-1.13)	1.15 (1.02-1.29)	1.13 (1.00-1.27)	
Dyspepsia –	1.15 (0.99-1.34)	1.29 (1.06-1.56)	1.10 (1.04-1.16)	1.04 (0.96-1.12)	1.25 (1.09-1.43)	1.01 (0.86-1.18)	
Constipation-	1.18 (1.00-1.39)	1.13 (0.91-1.41)	1.16 (1.09-1.24)	1.09 (1.01-1.19)	1.20 (1.03-1.40)	1.18 (1.00-1.40)	
Hearing loss –	1.11 (0.86-1.43)	1.05 (0.75-1.48)	1.12 (1.01-1.24)	0.94 (0.82-1.08)	1.29 (1.04-1.61)	0.94 (0.73-1.21)	
Diverticular disease of intestine	1.12 (1.01-1.24)	1.00 (0.87-1.14)	1.20 (1.15-1.24)	1.00 (0.96-1.05)	1.21 (1.11-1.32)	0.99 (0.90-1.09)	
Endometriosis -	1.26 (1.00-1.60)	1.14 (0.83-1.57)	1.20 (1.10-1.31)	1.30 (1.17-1.45)	0.95 (0.73-1.25)	1.31 (1.00-1.71)	
Meniere's disease-	1.32 (0.65-2.68)	2.49 (1.13-5.47)	1.51 (1.16-1.95)	1.07 (0.76-1.49)	1.25 (0.73-2.15)	1.05 (0.59-1.89)	

Fig. 4 Association of weight change from birth to childhood with individual conditions. Reference groups: changing to average childhood body size. Adjusted for age at baseline, sex, ethnicity, Townsend deprivation index, education level, smoking status, drinking status, physical activity, intake of fruits and vegetables, body mass index categories at baseline, maternal smoking around birth and breastfed as a baby. Diseases with hazard ratios>1 and P < 0.05 were marked in red, those with hazard ratios <1 and P < 0.05 were marked in blue.

childhood and its trajectories with the development of multimorbidity and chronic conditions.

When considering family history of several chronic diseases in the model, the associations of birth weight and childhood body size with multimorbidity were similar to the main results. Also, such association was consistent among participants with or without family history of most diseases, implying that the association of birth weight and childhood body size with multimorbidity may be independent of familial genetic factors. According to a Mendelian randomization study which provided evidence of genetic pleiotropy between birth weight and cardiometabolic risk factors, maternal intrauterine environment, in which maternal genetic variation influences offspring birth weight, was not the major predictor of offspring cardiometabolic outcomes, but birth weight determined by the individuals' own genetic factors was more likely to associate with cardiometabolic outcomes⁴⁵. This study, coupled with our results, provided converging evidence that familial genetic effects that predispose to offspring birth weight, were not the main determinant of some chronic diseases and multimorbidity in an individual's later life. Future studies using genetic data are warranted to verify such points.

The main strengths of this study include the large sample size, the long-term follow-up, the measurement of weight change, and the consideration of wide-range chronic conditions to measure multimorbidity using longitudinal data from UK Biobank. Limitations to this study also warrant consideration. First, the information on birth weight and childhood body size was selfreported by individuals, which was vulnerable to recall bias. Also, the childhood body size had no clear definition and was subjectively judged by individuals themselves, leading to the possibility of misclassification. Second, nearly half of participants recruited at baseline were excluded in our study, whose characteristics differed from the included, limiting the generalizability of our results. Third, the information on chronic conditions was linked to inpatient hospital data, missing out on the mild or undiagnosed forms of conditions, like depression and migraine. Fourth, the association of interest covered a large time span from birth to late-life, despite the inclusion of multiple covariates in our study, the potential for residual confounding of other factors could not been excluded. Fifth, although gestational age is an important factor affecting birth weight, we are not available to this data in UK Biobank. Last, we used family history of several major chronic disease as a proxy for genetic factors, future research utilizing genetic data are warranted to explore the interaction between genetic factors with birth weight and childhood body size in predicting multimorbidity.

In conclusion, birth weight, childhood body size and their changes are associated with higher risks of developing multimorbidity and many chronic diseases in late life. Early

ARTICLE

monitoring and interventions to keep a normal body size in childhood could have life-course benefits for preventing multimorbidity above and beyond individual chronic conditions.

Data availability

This research was conducted using the UK Biobank study under Application Number 66354. The original data for the study are available on the website: https://www.ukbiobank.ac.uk/. Detailed data on the demographics of study participants are shown in Supplementary Data 1 and Supplementary Data 2. All other data are available from the corresponding author (or other sources, as applicable) on reasonable request.

Code availability

The analytic code was shown in Supplementary Data 3.

Received: 12 April 2023; Accepted: 19 July 2023; Published online: 31 July 2023

References

- 1. WHO. *Global Nutrition Targets 2025: Anaemia Policy Brief* (World Health Organization; 2014).
- Ward, W. P. The Oxford Handbook of Economics and Human Biology (Oxford University Press, 2016).
- Liang, J. et al. Association between birth weight and risk of cardiovascular disease: evidence from UK Biobank. *Nutr. Metab. Cardiovasc. Dis.* 31, 2637–2643 (2021).
- Rashid, A., Agarwala, A., Novak, E. & Brown, D. L. Association of high birth weight with incident heart failure in the ARIC study. J. Am. Heart Assoc. 8, e011524 (2019).
- 5. Nehus, E. & Mitsnefes, M. Childhood obesity and the metabolic syndrome. *Pediatr. Clin. North Am.* 66, 31-43 (2019).
- Simmonds M., Llewellyn A., Owen C. G. & Woolacott N. Predicting adult obesity from childhood obesity: a systematic review and meta-analysis. *Obes. Rev.* https://doi.org/10.1111/obr.12334 (2016).
- Llewellyn, A., Simmonds, M., Owen, C. G. & Woolacott, N. Childhood obesity as a predictor of morbidity in adulthood: a systematic review and metaanalysis. *Obes. Rev.* 17, 56–67 (2016).
- Quek, Y.-H., Tam, W. W. S., Zhang, M. W. B. & Ho, R. C. M. Exploring the association between childhood and adolescent obesity and depression: a metaanalysis. *Obes. Rev.* 18, 742–754 (2017).
- Richardson, T. G. et al. Childhood body size directly increases type 1 diabetes risk based on a lifecourse Mendelian randomization approach. *Nat. Commun.* 13, 2337 (2022).
- Yeung, E. H., Zhang, C., Louis, G. M. B., Willett, W. C. & Hu, F. B. Childhood size and life course weight characteristics in association with the risk of incident type 2 diabetes. *Diabetes Care* 33, 1364–1369 (2010).
- Xu, X., Mishra, G. D. & Jones, M. Evidence on multimorbidity from definition to intervention: an overview of systematic reviews. *Ageing Res. Rev.* 37, 53–68 (2017).
- Adair, L. S. et al. Associations of linear growth and relative weight gain during early life with adult health and human capital in countries of low and middle income: findings from five birth cohort studies. *Lancet* 382, 525–534 (2013).
- Cai, Y., Shaheen, S. O., Hardy, R., Kuh, D. & Hansell, A. L. Birth weight, early childhood growth and lung function in middle to early old age: 1946 British birth cohort. *Thorax.* 71, 916–922 (2016).
- 14. Black, M. M. et al. Early childhood development coming of age: science through the life course. *Lancet* **389**, 77–90 (2017).
- Victora, C. G. et al. Maternal and child undernutrition: consequences for adult health and human capital. *Lancet* 371, 340–357 (2008).
- Rito, A. I. et al. Association between characteristics at birth, breastfeeding and obesity in 22 countries: the WHO European childhood obesity surveillance initiative—COSI 2015/2017. Obes. Facts 12, 226–243 (2019).
- Yan, J., Liu, L., Zhu, Y., Huang, G. & Wang, P. P. The association between breastfeeding and childhood obesity: a meta-analysis. *BMC Public Health.* 14, 1267 (2014).
- Patro Golab, B. et al. Influence of maternal obesity on the association between common pregnancy complications and risk of childhood obesity: an individual participant data meta-analysis. *Lancet Child Adoles. Health* 2, 812–821 (2018).
- Sudlow, C. et al. UK biobank: an open access resource for identifying the causes of a wide range of complex diseases of middle and old age. *PLoS Med.* 12, e1001779 (2015).

- 20. Macrosomia: ACOG Practice Bulletin Summary, number 216. Obstet. Gynecol.135, 246–248 (2020).
- Huang, X. et al. Birth weight and the risk of cardiovascular outcomes: a report from the large population-based UK biobank cohort study. *Front. Cardiovasc. Med.* 9, 827491 (2022).
- Doran T., Kontopantelis E., Reeves D., Sutton M. & Ryan A. M. Setting performance targets in pay for performance programmes: what can we learn from QOF? *BMJ* 348, g1595 (2014).
- Barnett, K. et al. Epidemiology of multimorbidity and implications for health care, research, and medical education: a cross-sectional study. *Lancet* 380, 37–43 (2012).
- Diederichs, C., Berger, K. & Bartels, D. B. The measurement of multiple chronic diseases-a systematic review on existing multimorbidity indices. J. Gerontol. A Biol. Sci. Med. Sci. 66, 301–311 (2011).
- 25. World Health Organization. *Health topics: Obesity and Overweight*. https:// www.who.int/news-room/fact-sheets/detail/obesity-and-overweight (2021).
- Zhang, H. et al. Familial factors, diet, and risk of cardiovascular disease: a cohort analysis of the UK Biobank. *Am. J. Clin. Nutr.* 114, 1837–1846 (2021).
- 27. Bradbury, K. E., Murphy, N. & Key, T. J. Diet and colorectal cancer in UK Biobank: a prospective study. *Int. J. Epidemiol.* **49**, 246–258 (2020).
- Zhang Y., et al. Association of dietary patterns and food groups intake with multimorbidity: a prospective cohort study. *Clin. Nutr.* https://doi.org/10. 1016/j.clnesp.2022.07.019 (2022).
- Sin, D. D. et al. The relationship between birth weight and childhood asthma: a population-based cohort study. *Arch. Pediatr. Adolesc. Med.* 158, 60–64 (2004).
- Yuan, W., Basso, O., Sorensen, H. T. & Olsen, J. Fetal growth and hospitalization with asthma during early childhood: a follow-up study in Denmark. *Int. J. Epidemiol.* 31, 1240–1245 (2002).
- Curhan, G. C. et al. Birth weight and adult hypertension, diabetes mellitus, and obesity in US men. *Circulation*. 94, 3246–3250 (1996).
- Yarbrough, D. E., Barrett-Connor, E., Kritz-Silverstein, D. & Wingard, D. L. Birth weight, adult weight, and girth as predictors of the metabolic syndrome in postmenopausal women: the Rancho Bernardo Study. *Diabetes Care.* 21, 1652–1658 (1998).
- Liao, L., Deng, Y. & Zhao, D. Association of low birth weight and premature birth with the risk of metabolic syndrome: a meta-analysis. *Front. Pediatr.* 8, 405 (2020).
- Cnattingius, S., Villamor, E., Lagerros, Y. T., Wikström, A. K. & Granath, F. High birth weight and obesity-a vicious circle across generations. *Int. J. Obes.* 36, 1320-1324 (2012).
- Wegelius, A. et al. An association between high birth weight and schizophrenia in a Finnish schizophrenia family study sample. *Psychiatry Res.* 190, 181–186 (2011).
- Ahlgren, M., Wohlfahrt, J., Olsen, L. W., Sørensen, T. I. A. & Melbye, M. Birth weight and risk of cancer. *Cancer.* 110, 412–419 (2007).
- Zhang, T. et al. Trajectories of childhood BMI and adult diabetes: the Bogalusa Heart Study. *Diabetologia*. 62, 70–77 (2019).
- Viner, R. M., Hindmarsh, P. C., Taylor, B. & Cole, T. J. Childhood body mass index (BMI), breastfeeding and risk of Type 1 diabetes: findings from a longitudinal national birth cohort. *Diabet. Med.* 25, 1056–1061 (2008).
- Lunde, A., Melve, K. K., Gjessing, H. K., Skjaerven, R. & Irgens, L. M. Genetic and environmental influences on birth weight, birth length, head circumference, and gestational age by use of population-based parentoffspring data. Am. J. Epidemiol. 165, 734–741 (2007).
- Barker, D. J. P. The origins of the developmental origins theory. J. Int. Med. 261, 412–417 (2007).
- Johnsson, I. W., Haglund, B., Ahlsson, F. & Gustafsson, J. A high birth weight is associated with increased risk of type 2 diabetes and obesity. *Pediatr. Obes.* 10, 77–83 (2015).
- 42. Fang, X. et al. Causal association of childhood obesity with cancer risk in adulthood: a Mendelian randomization study. *Int. J. Cancer* **149**, 1421–1425 (2021).
- Singhal, A. & Lucas, A. Early origins of cardiovascular disease: is there a unifying hypothesis? *Lancet* 363, 1642–1645 (2004).
- 44. Sun, J. et al. Weight change from childhood to adulthood and cardiovascular risk factors and outcomes in adulthood: a systematic review of the literature. *Obesity Rev.* **22**, e13138 (2021).
- Moen, G. H. et al. Mendelian randomization study of maternal influences on birthweight and future cardiometabolic risk in the HUNT cohort. *Nat Commun.* 11, 5404 (2020).

Acknowledgements

This work was funded by Zhejiang University. We thank all the lab members who gave support to the study and all the researchers and staff involved in UK Biobank.

Author contributions

All authors contributed to the study conception and design. X.X. contributed to the study conceptualization and supervised the whole project. Yu.Z. made the analysis plan, conducted the statistical analyses, and drafted and revised the manuscript. Ya.Z. verified the underlying data and drafted the initial manuscript. Y.C., R.M.C.-L., M.F., and S.C. provided support on the statistical methods and manuscript revision. All authors contributed to and approved the final manuscript. X.X. is the corresponding authors, had full access to all the data, and had final responsibility for the decision to submit for publication.

Competing interests

The authors declare no competing interests.

Additional information

Supplementary information The online version contains supplementary material available at https://doi.org/10.1038/s43856-023-00335-4.

Correspondence and requests for materials should be addressed to Xiaolin Xu.

Peer review information *Communications Medicine* thanks Steven Korzeniewski and Peter Nilsson for their contribution to the peer review of this work. A peer review file is available.

Reprints and permission information is available at http://www.nature.com/reprints

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this license, visit http://creativecommons.org/ licenses/by/4.0/.

© The Author(s) 2023